STUDY PROTOCOL



REVISED The CO-produced Psychosocial INtervention delivered

by GPs to young people after self-harm (COPING): protocol for

a feasibility study [version 2; peer review: 1 approved, 1

approved with reservations]

Faraz Mughal¹, Carolyn A. Chew-Graham¹, Benjamin Saunders¹, Sarah A. Lawton¹, Sarah Lewis¹, Jo Smith¹, Gillian Lancaster¹,

Ellen Townsend², Christopher J. Armitage¹³⁻⁵, Peter Bower⁶, Nav Kapur^{5,7,8},

David Kessler⁹, Alba X. Realpe^{10,11}, Nicola Wiles¹², Dennis Ougrin¹³, Martyn Lewis¹

¹School of Medicine, Keele University, Keele, England, UK

²School of Psychology, University of Nottingham, Nottingham, England, UK

³Manchester Centre for Health Psychology, School of Health Sciences, The University of Manchester, Manchester, England, UK ⁴Manchester University NHS Foundation Trust, Manchester Academic Health Sciences Centre, The University of Manchester, Manchester, England, UK

⁵NIHR Greater Manchester Patient Safety Research Collaboration, The University of Manchester, Manchester, England, UK ⁶Centre for Primary Care and Health Services Research, The University of Manchester, Manchester, England, UK

⁷Centre for Mental Health and Safety, Manchester Academic Health Sciences Centre, The University of Manchester, Manchester, England, UK

⁸Mersey Care NHS Foundation Trust, Liverpool, England, UK

⁹Centre for Academic Primary Care, Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, England, UK ¹⁰Population Health Sciences, University of Bristol, Bristol, England, UK

¹¹NIHR Bristol Biomedical Research Centre, University Hospitals Bristol and Weston NHS Foundation Trust, University of Bristol, Bristol, England, UK

¹²Centre for Academic Mental Health, Population Health Sciences, Bristol Medical School, University of Bristol, Bristol, England, UK ¹³Youth Resilience Unit, Centre for Psychiatry and Mental Health, Wolfson Institute of Population Health, WHO Collaborating Centre for Mental Health Services Development, Queen Mary University of London, London, England, UK

V2 First published: 08 May 2024, 4:27 https://doi.org/10.3310/nihropenres.13576.1 Latest published: 15 Oct 2024, 4:27 https://doi.org/10.3310/nihropenres.13576.2

Abstract

Background

Self-harm in young people is a growing concern and reducing rates a global priority. Rates of self-harm documented in general practice have been increasing for young people in the UK in the last two decades, especially in 13–16-year-olds. General practitioners (GPs) can intervene early after self-harm but there are no effective treatments

Open Peer Review				
Approval Status 🗹 ?				
	1	2		
version 2				
(revision)				
15 Oct 2024				
version 1	×	?		
08 May 2024	view	view		
1. Karen Rodham, University of Chichester,				

presently available. We developed the GP-led COPING intervention, in partnership with young people with lived experience and GPs, to be delivered to young people 16–25 years across two consultations. This study aims to examine the feasibility and acceptability of conducting a fully powered effectiveness trial of the COPING intervention in NHS general practice.

Methods

This will be a mixed-methods external non-randomised before-after single arm feasibility study in NHS general practices in the West Midlands, England. Patients aged 16–25 years who have self-harmed in the last 12 months will be eligible to receive COPING. Feasibility outcomes will be recruitment rates, intervention delivery, retention rates, and completion of follow-up outcome measures. All participants will receive COPING with a target sample of 31 with final follow-up data collection at six months from baseline. Clinical data such as selfharm repetition will be collected. A nested qualitative study and national survey of GPs will explore COPING acceptability, deliverability, implementation, and likelihood of contamination.

Discussion

Brief GP-led interventions for young people after self-harm are needed to address national guideline and policy recommendations. This study of the COPING intervention will assess whether a main trial is feasible.

Registration

ISRCTN (ISRCTN16572400; 28.11.2023).

Plain English Summary

Self-harm, usually by cutting or overdosing on tablets, affects young people, their families and friends, and society. Young people who selfharm are more likely to self-harm again and suffer from anxiety and depression. The leading cause of death in young people is suicide and over half of young people who die by suicide have previously selfharmed.

Self-harm results in around £128 million a year of costs to the NHS. The GP is the most frequently contacted health professional in the NHS for young people who have self-harmed. Therefore, the GP consultation provides a crucial opportunity to intervene early to reduce future self-harm. At present there are no interventions that work for GPs to use with young people who have self-harmed. We thus developed with GPs and patients a new GP-led brief treatment focusing on psychological and social factors for young people 16–25 years after self-harm (called COPING) to be delivered over two appointments. We would now like to test the COPING treatment in

Chichester, UK

2. **Maria Livanou**, King's College London, London, UK

Any reports and responses or comments on the article can be found at the end of the article.

general practice to understand whether a future large clinical trial of COPING is doable in the NHS.

We will recruit study sites from practices around England and train GPs at sites. We will identify potential participants through three recruitment strategies and all participants will receive the COPING treatment. We will collect follow-up data from young people at two, four, and six months from enrolment, and data will include information about mood, self-harm thoughts, and episodes of selfharm. GPs and participants will be invited for an interview to gain their experiences of COPING. An online survey of GPs in England will gauge interest about engagement in a future trial.

This study will assess whether COPING can be evaluated in a large clinical trial in NHS general practice. If feasible and acceptable there is the potential for COPING to substantially benefit patients, GPs, and the NHS.

Keywords

Protocol, self-harm, young people, general practice, general practitioner, feasibility, mixed methods

Corresponding author: Faraz Mughal (f.mughal@keele.ac.uk)

Author roles: Mughal F: Conceptualization, Funding Acquisition, Methodology, Project Administration, Writing – Original Draft Preparation, Writing – Review & Editing; Chew-Graham CA: Conceptualization, Methodology, Project Administration, Supervision, Writing – Review & Editing; Saunders B: Conceptualization, Methodology, Project Administration, Supervision, Writing – Review & Editing; Lawton SA: Conceptualization, Methodology, Project Administration, Writing – Review & Editing; Lewis S: Conceptualization, Methodology, Project Administration, Software, Writing – Review & Editing; Smith J: Conceptualization, Methodology, Project Administration, Software, Writing – Review & Editing; Lancaster G: Conceptualization, Methodology, Writing – Review & Editing; Townsend E: Conceptualization, Methodology, Writing – Review & Editing; Kapur N: Conceptualization, Methodology, Writing – Review & Editing; Kessler D: Conceptualization, Methodology, Writing – Review & Editing; Conceptualization, Methodology, Writing – Review & Editing; Wiles N: Methodology, Writing – Review & Editing; Ougrin D: Conceptualization, Methodology, Writing – Review & Editing; Lewis M: Conceptualization, Formal Analysis, Methodology, Supervision, Writing – Review & Editing

Competing interests: FM and NK were members of the 2022 self-harm NICE clinical guideline development committee. No other competing interests were disclosed.

Grant information: This project is funded by the National Institute for Health and Care Research (NIHR) under its ['NIHR Fellowships Programme' (Grant Reference Number NIHR300957) to Faraz Mughal]. CJA is supported by the NIHR Manchester Biomedical Research Centre and CJA and NK are supported by the NIHR Greater Manchester Patient Safety Research Collaboration (NIHR204295). CCG is funded by the NHS, NIHR or NIHR Applied Research Collaboration, West Midlands. The views expressed are those of the author(s) and not necessarily those of the NIHR or the Department of Health and Social Care.

The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Copyright: © 2024 Mughal F *et al.* This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

How to cite this article: Mughal F, Chew-Graham CA, Saunders B *et al.* The CO-produced Psychosocial INtervention delivered by GPs to young people after self-harm (COPING): protocol for a feasibility study [version 2; peer review: 1 approved, 1 approved with reservations] NIHR Open Research 2024, **4**:27 https://doi.org/10.3310/nihropenres.13576.2

First published: 08 May 2024, 4:27 https://doi.org/10.3310/nihropenres.13576.1

REVISED Amendments from Version 1

In the revised submission we have improved the Plain English summary; added some detail about recent trends in self-harm in young people in general practice; strengthened the introduction by stating the methods and functions of self-harm in young people and how self-harm varies with social circumstances; and described the patient and public involvement activity in more detail.

Any further responses from the reviewers can be found at the end of the article

Introduction

Background and rationale

Self-harm, defined as self-injury or self-poisoning irrespective of the purpose, impacts on patients, families, and health and care services^{1,2}. In young people, there is a lifetime self-harm prevalence of 26% in England, rising to 47% in those with a mental health disorder such as anxiety or depression^{3,4}. The annual cost of hospital treated self-harm in the National Health Service (NHS) is estimated to be £128 million⁵.

Self-harm is closely associated with death by suicide, and in young people self-harm is a major risk for premature death with around 40 years of life lost to 'external-cause' death^{6,7}. Methods of self-harm in young people include cutting with a sharp object, overdosing on medications, skin pinching or scratching, burning, and hair pulling⁸. The functions of self-harm in young people aged 16–25 specifically are handling emotional states; self-punishment; coping with mental illness and trauma; and positive thoughts and protection⁹. Young people are more likely to self-harm if their family is less affluent and they live in one parent households¹⁰. Rates of self-harm in young people coded in general practice health records have increased over the last two decades, and specifically for 13- to 16-year-old females from March 2020 to March 2022^{11–13}.

General practitioners (GPs) are often the first point of healthcare access for young people after self-harm and young people who present to general practice after self-harm are 17 times more likely to die by suicide⁷. GPs are well placed and have some skills to manage self-harm early in primary care: practice nurses have reported not feeling competent, and NHS Talking Therapies do not offer therapy for self-harm^{14,15}. However, at present, there are no effective treatments that GPs can offer young people after self-harm¹⁶.

Reducing self-harm in young people is an international priority². There is evidence psychosocial treatments, such as cognitive behaviour therapy-based psychotherapy and dialectical behaviour therapy, can reduce repeat self-harm episodes in young people but these were delivered in non-general practice settings^{17,18}. To address this intervention evidence gap, we developed using a combination complex intervention development approach (evidence and theory approach and a partnership approach with young people with lived experience of self-harm and GPs) a brief GP-led psychosocial intervention

(COPING) to help young people aged 16–25 years, across two GP consultations, avoid future self-harm (COPING-ID study, IRAS 294180).

Through early intervention in general practice there is potential to prevent long-term self-harm in young people, reduce NHS self-harm treatment costs, and future mental health and specialist (emergency, paediatric, medical) care costs. If the COPING intervention is feasible and acceptable, a future main randomised controlled trial (RCT) has the potential to bring substantial benefit to patients and the NHS. This feasibility study aims to examine aspects of the feasibility of future pragmatic RCT to assess the effectiveness of COPING in the NHS and inform its design.

Objectives

To examine aspects of the feasibility of a fully powered RCT of the COPING intervention across the areas of recruitment, delivery, and retention and follow-up. Specific objectives are:

- i) Estimate participant consent and willingness of enrolled participants and GPs to be randomised in a future RCT
- ii) Assess GP fidelity to COPING delivery and participant adherence to COPING
- iii) Examine completeness of participant follow-up data collection
- iv) Explore the acceptability of COPING for both participants and GPs, and deliverability of COPING for GPs

Methods

Study design

This study is a mixed-methods external non-randomised before-after single arm feasibility study.

This study is registered on ISRCTN registry, 28.11.2023: ISRCTN16572400; https://www.isrctn.com/ISRCTN16572400.

Study setting

NHS general practices in the West Midlands, England.

Eligibility criteria

Patients will be eligible to participate if they are aged 16–25 years and have self-harmed (according to the definition of the National Institute for Health and Care Excellence) in the last 12 months¹. Exclusion criteria include acute risk of suicide or harm to others necessitating urgent referral/admission; moderate to severe learning disabilities; current psychological therapy; patient inappropriate for COPING at clinician discretion; and lacking capacity for informed consent.

At general practice study sites, GPs will be eligible to deliver COPING if they complete the site COPING training.

Intervention

All participants will receive the COPING intervention delivered by GPs to young people aged 16–25 years after self-harm across two GP appointments. COPING's development was guided by the new Medical Research Council's (MRC) framework for developing and evaluating complex interventions¹⁹. COPING was developed through a 'combination' development approach (O'Cathain's taxonomy - integrating both theory and evidence-based and partnership approaches)²⁰. It was informed by the Behaviour Change Wheel, and included coproduction and prototyping (Hawkin's framework) in its design and refinement^{21,22}.

COPING comprises eight behaviour change techniques (BCTs) aimed at influencing the selected target behaviours for the young person and the GP. The target behaviours are: i) the young person using new skills to avoid self-harm and ii) encouraging GP use of COPING in usual NHS practice. The BCTs are problem solving, self-monitoring of behaviour, goal setting behaviour, review behaviour goal, social support (practical), instruction on how to perform the behaviour, demonstration of the behaviour, and rewarding completion. In the first COPING appointment the GP and participant agree on new skills aligned to the function of the self-harm to use when urges of self-harm occur, and a diary is provided. In the second appointment the GP reviews progress with the participant. The first appointment will be in-person and the format and time (from two to four weeks) of the second COPING appointment will be decided between the GP and participant.

The COPING training will be delivered by the Chief Investigator (CI) to site GPs in person and last up to 60 minutes. The training will include presenting the study background (including COPING-ID), rationale and aim, providing the COPING intervention materials, offering to show an example COPING delivery video, and highlighting study procedures and safety reporting.

Outcomes

The primary outcomes are:

- Consent to study of eligible patients
- Young people participants willing to be randomised in a main RCT
- Fidelity to COPING
- Participant adherence to COPING
- Participant follow-up rate of outcome measures
- Acceptability of COPING to participants and GPs
- Deliverability of COPING

The secondary outcomes are:

- Response rates of invited practices to participate
- Identification of patients at sites

- Proportion of identified patients eligible for study inclusion
- Participant recruitment uptake through recruitment strategies
- Willingness of GPs to be randomised in a main RCT
- Assessment of outcome measure to use as the primary outcome in a main RCT
- Proportion of participants with repetition of self-harm
- Participant attrition, follow-up, and withdrawal rates

This feasibility study is not designed to evaluate health outcome measures formally. We will collect clinical effectiveness data that may support the justification for a future main trial. These data may also be used to decide which outcomes could and should be collected in a future RCT, and to estimate parameters required to calculate the sample size calculation for a fully powered RCT.

The progression decision-making process will include consideration of the feasibility study progression outcomes (aligned to the primary outcomes: see Table 1) with the study steering committee (SSC), patient and public involvement (PPI) group, and research team to consider if moving to a fully powered RCT is feasible.

Participant timeline

The schedule of study events is listed in Table 2.

Sample size

Non-randomised COPING delivery:

We will aim to recruit approximately six general practice sites and around 3 GPs at each site. The incidence rate of recorded self-harm in young people in general practice is around 60 per 10,000 person-years²³. This equates to 60 per 1,000 for the 16-25 age group over one year. If half are eligible for the study (30 per 1,000) and two-thirds of those are identified by our recruiting methods (20 per 1,000) then in a typical GP practice (assumed 6,000 registered patients of which 900 (15%) are in the 16-25 age group) about 18 eligible young people will contact their GP practice per year after self-harm (1.5 eligible young people per practice per month). If six practices are recruited to take part, nine young people would be expected to be eligible per month, and around 104 (the target eligible population size) over about 12 months. From this number (n=104) we would expect to recruit 31 participants (30%) (fulfilling the Go signal for progression for the recruitment criterion - see Table 1) into the study.

The sample size of n=31 participants is required to test against the feasibility progression criteria (specifically the cut-off for a STOP signal) with at least 90% power, using the normal approximation method for binary outcomes using a 1-sample 1-sided test with 5% alpha (type I error) using the approach of Lewis *et al.*²⁴ In line with the recommendation of the CONSORT-extension for randomised pilot and feasibility

Criteria	Go – proceed to RCT	Amend – proceed with minor changes	Stop – do not proceed unless major changes are possible
Consent to study in eligible patients	>30%	15–30%	<15%
Young people participants willing to be randomised in a future RCT	>85%	60-85%	<60%
Fidelity to COPING treatment checklist	>67%	40-67%	<40%
Participant adherence to COPING	>67%	40-67%	<40%
Participant follow-up rate of outcome measures	>80%	50-80%	<50%
Acceptability of COPING to GPs and patients	COPING judged very acceptable by GPs and patients by qualitative data	COPING judged acceptable by GPs and patients by qualitative data	COPING judged somewhat acceptable by GPs and patients by qualitative data
Delivery of COPING by GPs	Delivery of COPING judged strongly feasible by qualitative data	Delivery of COPING judged feasible by qualitative data	Delivery of COPING judged possibly feasible by qualitative data

Table 1. Progression criteria for a future main trial.

studies this provides necessary justification for the sample size requirement for the evaluation of the progression criteria for this feasibility study²⁵.

GP survey:

A sample of 100 GPs across England will be surveyed to elicit GP views about a future main RCT of COPING. Achieving 100 GP responses from a convenience sample has recently been shown to be achievable²⁶. If we expect 50% to respond then n=50 would be sufficient to provide a margin of error no greater than 0.1 standard error (SE) around an estimate for a proportion based on a binary response item "e.g., would you be willing to participate in a future RCT of COPING (yes/ no)". Therefore, if 25/50 respond 'yes', it will generate an estimate of the proportion who answer positively of 0.5 (50%) with margin of error <= +/-0.1 relating to a z-value of +/-1.282 allowing a 90% 1-sided confidence interval (CI) and 80% 2-sided CI (providing sufficiently precise confidence interval estimation) around the observed estimate.

Nested qualitative study:

It is anticipated that a sample of around 16–20 participants (8–10 young people participants and 8–10 GPs) is likely enough to identify problems in COPING delivery and achieve data saturation, where data no longer offers new insights²⁷.

Recruitment

General practices in the National Institute for Health and Care Research Local Clinical Research Network (CRN) West Midlands will be invited to participate in the study by email. Practices who agree to participate will receive a local information pack, a site initiation visit, COPING training, and an investigator site file. Recruited practices therefore become study sites. Each site will have a designated principal investigator (PI), and each site PI will identify GPs who want to deliver COPING to participants. GPs who use COPING with participants become study co-investigators

Patient recruitment to self-harm clinical trials in NHS general practice is untested and thus we will test three participant recruitment strategies which have been informed by the study's PPI group:

- 1) General practice electronic healthcare records: practice database searches will be developed for sites to use to identify potentially eligible patients.
- Study advertisement: recruitment posters will be displayed in site waiting rooms, linked community pharmacies, and nearby educational institutions to share the study opportunity.
- Routine clinical practice: members of the clinical team can signpost potentially eligible patients to the study opportunity during routine consultations.

In strategy one, potential eligible patients will be identified at research sites through searching practice-based electronic healthcare registers using a standardised strategy. A practice clinician will screen the identified list of potentially eligible patients and exclude those ineligible. The remaining eligible patients, with a prior telephone check where appropriate, will be sent a text message and postal invitation from the practice signposting them to the study webpage, where the participant information sheet (PIS) and e-consent form can be found.

Schedule of events		Enrolment	Baseline data collection	COPING Follow-up		o		
		0 weeks	0 weeks	T1	T2	8 weeks	4 months	6 months
Enrolment								
	Eligibility screen	1						
	Informed consent	1						
Participant descriptors								
Demographics	DOB, sex, gender		\checkmark					
Language	Main language		\checkmark					
Ethnicity	Ethnic origin		\checkmark					
Education status	Туре		✓					
Employment status			\checkmark					
Internet use and access	Frequency and method		1					
Medications								
Prescribed medications	Medication name, dosage, and frequency		1					
Clinical outcomes								
Past self-harm	Method, timing of last self-harm, and frequency of episodes		1					
Mood and suicidal or self- harm thoughts	PHQ-9		1			1	1	1
Attempted suicide	SBQ-R		\checkmark			1	1	\checkmark
COPING delivery								
Duration	Length of consultations (minutes)			1	1			
Deciding skills				✓	1			
Goal setting				1	1			
Provision of COPING diary				1				
Attendance at 2 nd COPING consult					1			
Review of diary					1			
Interviews								
Adverse events								
Recording of adverse events	Classifying and logging adverse events			•				

Table 2. Schedule of study events.

 $\rm T_1$ – first COPING consult, $\rm T_2$ – second COPING consult

A reminder will be sent after two weeks if no response. A simplified two-page PIS summarising what will be required from participants enrolled in the study will be made available on the study website.

Other potentially eligible patients may also be approached opportunistically by a member of the clinical care team in routine practice (strategy three). These patients and those who see the study advertised in the community around participating A convenience sample of GPs around England will be sought to complete an online survey. GP participants will be recruited through the professional networks and social media. GPs will have the option to be entered into a random draw for two £25 e-vouchers on completion of the survey. Young people participants who have given consent for interviews will be purposively sampled across practice location and deprivation and GPs who have delivered COPING at sites will be invited for interview and asked to e-consent. Both will be reimbursed for their time on completion of interviews.

for the study, they will be invited to consent into the study.

Patient and Public Involvement

The design of this study was informed by PPI through a self-harm in young people PPI advisory group, including young people and carers, led by the CI at Keele University, and by consulting the McPin Foundation's Young People's Advisory Group and the Kings College London/NIHR Maudsley Biomedical Research Centre Young Person's Mental Health Advisory Group. This feasibility study will be supported by the COPING study PPI advisory group consisting of five young people who self-harm and three carers of young people who self-harm. This group contributed to the design of COPING by generating ideas for consideration in co-production activities in the COPING-ID study.

The first study meeting occurred on 9 May 2023 where all members of the advisory group attended, and the group helped design the participant eligibility and study progression criteria. Members also gave feedback about the content and design of the participant study information sheet and participant recruitment poster. The group will meet two monthly to support the delivery and dissemination of this study.

Data collection methods

Patients who consent to participate will be asked to complete an online baseline questionnaire which will collect data on participant age, sex, ethnicity, medication, number and time of recent self-harm episodes, the patient health questionnaire (PHQ-9), the Suicidal Behaviours Questionnaire-Revised (SBQ-R), and educational status.

Participant follow-up outcome data collection will be at eight weeks, four, and six months, from baseline using a self-report electronic questionnaire. Data collected will include individual participant repetition of self-harm via the SBQ-R, and PHQ-9 scores will be attained at these timepoints. Participants will be asked if they would be willing to be randomised in a future RCT of COPING in the eight-week follow-up questionnaire. One email or SMS text message reminder will be sent after one week if there has been no response after each questionnaire invitation. If a participant withdraws from the study, their data will be retained and used for the study. In the situation a participant withdraws from completing the COPING treatment, they will still be invited to complete the follow-up questionnaires. GPs will record training time, length of COPING consultations, and attendance of participant at 2nd COPING consult on Case Report Forms (CRF). General practice medical records of participants will be reviewed and relevant aspects extracted and depersonalised at participant sites, and securely transferred to the study team to obtain information for example about past medical history and prescriptions.

GPs around England will be invited to complete an online survey. This survey will ask GPs about the number of selfharm presentations in young people 16–25 years they see each month; willingness to be involved in a future RCT, including being randomised; barriers and enablers to COPING implementation; and likelihood of contamination in a future individually randomised RCT.

In the nested qualitative study, semi-structured interviews with participants and GPs will explore acceptability of COPING and acceptability of COPING consults being recorded in a future trial, potential unintended harms, willingness to take part in a future trial and to be randomised, 'what is usual care', barriers and enablers for COPING implementation, and views on outcome selection and measurement. The interview topic guide will be sensitised by the Theoretical Framework of Acceptability and Theoretical Domains Framework and iteratively revised^{28,29}. FM will conduct interviews face-to-face at sites or remotely. Interview participants will be able to withdraw their data up to one week after the interview.

Data management

Data will be collected in electronic and paper form. Data from CRFs at sites will be stored on secure Keele University electronic servers. Participant self-report data will be collected electronically via the Keele REDCap and likely facilitated by Keele Health Survey. Transfer of data between research sites and the COPING study team will be conducted via secure email transfer.

Interviews will be audio-recorded and audio files immediately transferred onto Keele's secure network. Audio files will be securely shared with The Transcription Company, a thirdparty company with whom Keele University have established data protection agreements. A dedicated password restricted study database will be developed and maintained on Keele University servers, managed by the Keele University Information & Data Security department, and will be the final repository for the data collection. Each participant will be allocated a unique study number, so that only anonymised data are used for analysis. Depersonalised medical records will be linked to a participant's unique study number and other participant data collected.

All data will be stored on Keele University storage services within the UK and protected by industry standard security tools. All confidentiality arrangements adhere to relevant data protection regulations and guidelines (Data Protection Act 2018, UK GDPR, Caldicott, General Medical Council (GMC), Medical Research Council (MRC) UK Policy), the Confidentiality NHS Code of Practice, and the CI (FM) and Data Custodian have responsibility to ensure the integrity of the data and that all confidentiality procedures are followed.

Statistical methods

Participant characteristics, eligibility and recruitment rates, attendance at second COPING consult, baseline and follow-up questionnaire response rates, GP survey responses, and resource use will be reported descriptively using SPSS (Statistical Package for the Social Sciences) v27 or later³⁰. Continuous variables (e.g., age) will be reported with means and standard deviations or median (IQR) as appropriate and categorical variables (e.g., sex) will be reported with frequencies and percentages.

Analysis of outcomes against the predefined feasibility study progression criteria will include estimated percentages with 95% confidence intervals (e.g., rates of consent or willingness to be randomised). Participant follow-up rate of outcome measures will be assessed as the total completions attained of the SBQ-R measure at all three timepoints as a percentage of the possible denominator (n=93). Participant adherence to the COPING intervention will be ascertained by examining the proportion of participants who attend for the 2nd COPING consult. Participant PHQ-9 and SBQ-R scores at baseline and follow-up timepoints will be compared using paired t-test (if normal distribution) or Wilcoxon paired rank test³¹.

Fidelity of the COPING intervention will be measured retrospectively by reviewing CRF data completed by site GPs against the COPING delivery checklist. Overall fidelity will be calculated as a percentage of the participants where the number of COPING items successfully delivered is at least three out of five (from the five BCTs targeting the young person's target behaviour), as agreed with the SSC.

We will investigate how much missing data there is in this study to inform the planning of a future RCT. This will occur at two levels: i) observe how much loss to follow-up there will be (participant attrition or withdrawal), and ii) assess completion rates, including missing data for specific data items, of follow-up questionnaires.

Qualitative analysis

Interview data about the acceptability and deliverability of COPING, and free text responses from the survey, will be analysed by practical thematic analysis and themes mapped to both the Theoretical Domains Framework and Theoretical Framework of Acceptability to support the identification of barriers and facilitators to COPING implementation in the NHS and COPING acceptability^{28,29,32}. The CI (FM) will lead analysis and it will be facilitated by NVivo 12 or later³³.

Integration of data

Quantitative and qualitative data will be integrated using a triangulation protocol to enhance the validity of findings and assess data complementarity, convergence, and dissonance across key findings³⁴.

Data monitoring and auditing

The CI will manage this study day-to-day and will convene a Study Management Group (SMG) who will be members of the research team to support study delivery. The SMG will be responsible for monitoring study progress and conduct, study protocol implementation, and that the safety and wellbeing of participants is maintained where appropriate. The SMG will escalate any concerns to the SSC and sponsor as required. The CI will meet the SMG at monthly study meetings with communication facilitated by email where needed.

The SSC, which includes an independent statistician, a GP clinical trialist, and a GP researcher, have approved the finalised study protocol, agreed meeting schedules, understand reporting lines to the study sponsor and funder, and agree to the SSC terms of reference. The role of the SSC is to ensure the study is delivered ethically and safely, and to consider the study's progress in light of new information. The SSC is independent from the sponsor and members have declared no conflicts of interest. The SSC will meet with the CI and some members of the SMG at least once a year. Through the chair of the SSC, the SSC will make recommendations to the CI, SMG, and sponsor, and if early study termination is felt appropriate, the chair will inform the study sponsor.

Authorised representatives of the study sponsor will have access to study data for monitoring and auditing purposes. The CI will conduct audits across the research study following a risk-proportionate approach.

Harms

The CI will monitor and log all adverse events (AEs) and serious adverse events (SAEs), including their assessment of expectedness (and relatedness where needed), in accordance with the protocol which is in agreement with Keele University Health and Social Care standard operating procedure RM16 Safety Reporting for non-CTIMP research using the study's CRFs for AEs and SAEs. At site initiation visits, site PIs will be appropriately trained and made aware of this study's safety reporting requirements. Where a SAE is identified at sites, it must be reported to the study team within 24 hours. Due to the nature of this study, engagement in acts of self-harm from participants during the study is to be expected.

Ethics and dissemination

This study has received ethics approval from NHS East of England - Cambridge East Research Ethics Committee on 13.11.2023 (23/EE/0238) and Health Research Authority and Health and Care Research Wales Approval on 14.11.2023 (IRAS 327529).

There is risk that participants may describe self-harm or suicidal intent when returning self-report questionnaires or while partaking in an interview. Two risk protocols have been developed. Should likely risk to participants be identified, the relevant risk protocol will be activated, and the participant's GP notified. The protocols outline when breaking confidentiality may be considered. The CI is a GP experienced in assessing risk of self-harm or suicide in patients and all participants will receive a list of free support services for use when needed.

The final full study report will be available from the CI, and findings will be prepared for submission to peer-reviewed journals. All publications, presentations, and correspondence generated will acknowledge NIHR as the funding source. Authorship will be determined in accordance with The International Committee of Medical Journal Editors criteria, and other contributors acknowledged, on manuscripts submitted for peer-reviewed publication.

The results of the study will be available to all stakeholders in ways that are easy to access at no cost. The PPI group will advise on how to best design and disseminate results to the wider public. The findings will also be published on the COPING study website. Avenues of dissemination include open-access peer-reviewed publications in academic journals, infographics for social media, presentations at clinical, public, and academic conferences, and NHS, NIHR, professional and public networks.

Discussion

This study will determine the feasibility of conducting a fully powered effectiveness RCT of the COPING intervention in the NHS. The COPING intervention is a novel GP-led intervention that aims to address the intervention evidence gap for young people after self-harm. It was designed through coproduction with patients and GPs and its implementation in usual NHS general practice was considered from the start.

This feasibility study is designed to be pragmatic and deliverable in usual NHS general practice and has been informed by PPI. Three recruitment strategies will be tested to enhance understanding about identifying potentially eligible participants and the study's progression criteria will determine whether a future main RCT is justified. Qualitative data will provide insights into how to optimise COPING and improve its delivery in practice. A national survey of GPs will provide data to help consider the design of an RCT. The integration of different types of data will enhance the credibility of findings.

The 2023–2028 suicide prevention in England strategy states that a priority area is for tailored support to be available to priority groups³⁵. Young people who have self-harmed are a high-risk priority group (encompassing two priority groups: young people and people who have self-harmed) and therefore if acceptable, deliverable, and feasible to be tested in a fully powered RCT, COPING has the potential to benefit patients, GPs, and the NHS, and address this national priority area.

Ethics and consent

This study has received ethics approval from NHS East of England - Cambridge East Research Ethics Committee on 13.11.2023 (23/EE/0238) and Health Research Authority and Health and Care Research Wales Approval on 14.11.2023 (IRAS 327529). Participants will provide written informed consent via the study's e-consent form.

Data availability

Underlying data

No data are associated with this manuscript.

Reporting guidelines

This manuscript has been written to include the recommended items from the SPIRIT 2013 checklist for interventional trials.

References

- National Institute for Health and Care Excellence: Self-harm: assessment, management and preventing recurrence. [NG225], 2022. Reference Source
- Knipe D, Padmanathan P, Newton-Howes G, et al.: Suicide and self-harm. Lancet. 2022; 399(10338): 1903–16.
 PubMed Abstract | Publisher Full Text
- McManus S, Gunnell D, Cooper C, et al.: Prevalence of non-suicidal self-harm and service contact in England, 2000-14: repeated cross-sectional surveys of the general population. Lancet Psychiatry. 2019; 6(7): 573–581. PubMed Abstract | Publisher Full Text | Free Full Text
- NHS Digital: Mental health of children and young people in England, 2017. [PAS]. 2018.
 Reference Source
- Tsiachristas A, Geulayov G, Casey D, et al.: Incidence and general hospital costs of self-harm across England: estimates based on the multicentre study of self-harm. Epidemiol Psychiatr Sci. 2020; 29: e108.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Bergen H, Hawton K, Waters K, et al.: Premature death after self-harm: a multicentre cohort study. Lancet. 2012; 380(9853): 1568–74.
 PubMed Abstract | Publisher Full Text

- Morgan C, Webb RT, Carr MJ, et al.: Incidence, clinical management, and mortality risk following self harm among children and adolescents: cohort study in primary care. BMJ. 2017; 359: j4351.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Mughal F, Ougrin D, Stephens L, *et al.*: Assessment and management of selfharm and suicide risk in young people. *BMJ*. 2024; 386: e073515.
 PubMed Abstract | Publisher Full Text
- Mughal F, Chew-Graham CA, Babatunde OO, et al.: The functions of self-harm in young people and their perspectives about future general practitionerled care: a qualitative study. *Health Expect*. 2023; 26(3): 1180–1188.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Public Health England: Intentional self-harm in adolescence: an analysis of data from the Health Behaviour in School-aged Children (HBSC) survey for England, 2014. 2017. Reference Source
- Carr MJ, Ashcroft DM, Kontopantelis E, *et al.*: The epidemiology of self-harm in a UK-wide primary care patient cohort, 2001-2013. *BMC Psychiatry*. 2016; 16: 53.
 PubMed Abstract | Publisher Full Text | Free Full Text
- 12. Morgan C, Webb RT, Carr MJ, et al.: Incidence, clinical management, and

mortality risk following self harm among children and adolescents: cohort study in primary care. *BMJ.* 2017; **359**: j4351. PubMed Abstract | Publisher Full Text | Free Full Text

- 13. Trafford AM, Carr MJ, Ashcroft DM, *et al.*: **Temporal trends in eating disorder** and self-harm incidence rates among adolescents and young adults in the UK in the 2 years since onset of the COVID-19 pandemic: a populationbased study. *Lancet Child Adolesc Health*. 2023; 7(8): 544–554. PubMed Abstract | Publisher Full Text
- Bailey D, Kemp L, Wright N, et al.: Talk About Self-Harm (TASH): participatory action research with young people, GPs and practice nurses to explore how the experiences of young people who self-harm could be improved in GP surgeries. Fam Pract. 2019; 36(5): 621–626. PubMed Abstract | Publisher Full Text
- Mughal F, Troya MI, Dikomitis L, *et al.*: Role of the GP in the management of patients with self-harm behaviour: a systematic review. *Br J Gen Pract.* 2020; 70(694): e364–e373.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Mughal F, Clarke L, Connolly R, *et al.*: Improving the management of selfharm in primary care. *Br J Gen Proct*. England, 2023; 73(729): 148–9.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Witt KG, Hetrick SE, Rajaram G, et al.: Interventions for self-harm in children and adolescents. Cochrane Database Syst Rev. 2021; 3(3): CD013667. PubMed Abstract | Publisher Full Text | Free Full Text
- Witt KG, Hetrick SE, Rajaram G, et al.: Psychosocial interventions for selfharm in adults. Cochrane Database Syst Rev. 2021; 4(4): CD013668.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Skivington K, Matthews L, Simpson SA, et al.: A new framework for developing and evaluating complex interventions: update of Medical Research Council guidance. BMJ. 2021; 374: n2061.
 PubMed Abstract | Publisher Full Text | Free Full Text
- O'Cathain A, Croot L, Sworn K, et al.: Taxonomy of approaches to developing interventions to improve health: a systematic methods overview. *Pilot Feasibility Stud.* 2019; 5: 41.
- PubMed Abstract | Publisher Full Text | Free Full Text

 21.
 Michie S, Atkins L, West R: The Behaviour Change Wheel. Great Britain: Silverback; 2014.
 - Reference Source
- Hawkins J, Madden K, Fletcher A, et al.: Development of a framework for the co-production and prototyping of public health interventions. *BMC Public Health*. 2017; 17(1): 689.
 PubMed Abstract | Publisher Full Text | Free Full Text
- 23. Cybulski L, Ashcroft DM, Carr MJ, et al.: Temporal trends in annual incidence

rates for psychiatric disorders and self-harm among children and adolescents in the UK, 2003-2018. BMC Psychiatry. 2021; 21(1): 229. PubMed Abstract | Publisher Full Text | Free Full Text

- Lewis M, Bromley K, Sutton CJ, et al.: Determining sample size for progression criteria for pragmatic pilot RCTs: the hypothesis test strikes back! *Pilot Feasibility Stud.* 2021; 7(1): 40.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Eldridge SM, Chan CL, Campbell MJ, et al.: CONSORT 2010 statement: extension to randomised pilot and feasibility trials. BMJ. 2016; 355: i5239. PubMed Abstract | Publisher Full Text | Free Full Text
- Jones OT, Jurascheck LC, Utukuri M, et al.: Dermoscopy use in UK primary care: a survey of GPs with a special interest in dermatology. J Eur Acad Dermatol Venereol. 2019; 33(9): 1706–12.
 PubMed Abstract | Publisher Full Text | Free Full Text
- O'Cathain A, Hoddinott P, Lewin S, et al.: Maximising the impact of qualitative research in feasibility studies for randomised controlled trials: guidance for researchers. Pilot Feasibility Stud. 2015; 1(1): 32.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Sekhon M, Cartwright M, Francis JJ: Acceptability of healthcare interventions: an overview of reviews and development of a theoretical framework. BMC Health Serv Res. 2017; 17(1): 88.
 PubMed Abstract | Publisher Full Text | Free Full Text
- Cane J, O'Connor D, Michie S: Validation of the theoretical domains framework for use in behaviour change and implementation research. Implement Sci. 2012; 7: 37.
 PubMed Abstract | Publisher Full Text | Free Full Text
- 30. Corp IBM: **IBM SPSS statistics**. 27 ed, 2020. **Reference Source**
- Siegel S: Nonparametric statistics for the behavioral sciences. McGraw-Hill; 1956.
 Reference Source
- Saunders CH, Sierpe A, von Plessen C, et al.: Practical thematic analysis: a guide for multidisciplinary health services research teams engaging in qualitative analysis. BMJ. 2023; 381: e074256. PubMed Abstract | Publisher Full Text
- 33. QSR International Pty Ltd: NVivo (Version 12). 2018.
- O'Cathain A, Murphy E, Nicholl J: Three techniques for integrating data in mixed methods studies. *BMJ*. 2010; 341: c4587.
 PubMed Abstract | Publisher Full Text
- Department of Health and Social Care: Suicide prevention in England: 5 year cross-sector strategy. 2023. Reference Source

Open Peer Review

Current Peer Review Status: 💙 🏅

Version 1

Reviewer Report 23 August 2024

https://doi.org/10.3310/nihropenres.14737.r31764

© **2024 Livanou M.** This is an open access peer review report distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

-		
	•	
	Ξ.	
	•	

Maria Livanou

¹ Department of Psychology, King's College London, London, England, UK ² Department of Psychology, King's College London, London, England, UK

I would like to thank the authors for this important work, given the alarmingly high numbers of young people who self-harm. This study protocol outlines the procedures involved in testing the feasibility of the COPING intervention, designed to be delivered by general practitioners (GPs) over two consultations to reduce self-harming in young people aged 16-25 years. COPING was coproduced with young people with lived experience and GPs. The present feasibility trial aims to examine the acceptability of conducting a fully powered trial.

Abstract

Background

What about the increasing numbers in the UK? Any specific trends that could be reported here to showcase the magnitude of the problem?

Methods

• It would help if the authors were specific about the number of GP practices involved in the feasibility study.

Discussion

- "Brief GP-led interventions for young people after self-harm are needed and address national guideline and policy recommendations". Do the authors mean to address?
- Any expected outcomes?

Plain English Summary

"Young people who self-harm experience further self-harm, anxiety and depression".

Do the authors mean that young people who self-harm are more likely to repeat such behaviours in the future?

The costs of self-harming would be a better fit in the second paragraph as the first one presents and describes the problem of self-harming. Here the consequences and impact on the young people and their families would be more relevant.

Introduction

The authors mention that 47% of those who self-harm have a mental illness. It would help if they were specific about which ones are related to self-harming. E.g., depression, eating disorders.

The types of self-harming should be addressed in more depth and, especially some forms not widely known such as picking skin, hair pulling, which are very prevalent in young people. Then, it is important to discuss why young people might be engaging in self-harming behaviours.

The introduction would benefit from including specific trends in self-harming with regards to demographic characteristics (e.g., age groups, gender, ethnicity) and clinical features (e.g., depression, anxiety, eating disorders). For example, young people living in deprived areas, those with chronic mental health conditions such as anorexia nervosa, those who have lost a parent, and those in foster care might be presenting with a higher risk for self-harming. I think such social determinants are crucial when looking at self-harming in young people.

The authors refer to GPs as having some skills to manage self-harm so I wonder about training in GPs in detecting and exploring self-harm. Do they receive adequate training to address and cope with such issues?

Talking therapies might not be offering specific targeted interventions but self-harm is a symptom so the other risk factors in a young person's life may be addressed.

It would be helpful to explore more what elements of existing evidence-based interventions and approaches have informed the current intervention. What is the framework other than co-production? The co-production process could be described to show how this led to the development of COPING.

Methods

Eligibility criteria

How will the authors define self-harming?

I wonder what is the role of carers in the process e.g., parents and how they can be involved at least in the qualitative phase of the study.

A table outlining the design and format of the intervention would be helpful and also the 8 behaviour change techniques and how these will be applied across the two consultations.

Patient and Public Involvement

This section could include further information about the Steering Groups and how the content of the intervention was developed.

Is the rationale for, and objectives of, the study clearly described?

Yes

Is the study design appropriate for the research question? Yes

Are sufficient details of the methods provided to allow replication by others? Partly

Are the datasets clearly presented in a useable and accessible format? Partly

Competing Interests: No competing interests were disclosed.

Reviewer Expertise: child and adolescent mental health

I confirm that I have read this submission and believe that I have an appropriate level of expertise to confirm that it is of an acceptable scientific standard, however I have significant reservations, as outlined above.

Author Response 10 Oct 2024

Faraz Mughal

We thank the reviewer for their positive comment on our manuscript and work.

What about the increasing numbers in the UK? Any specific trends that could be reported here to showcase the magnitude of the problem?

We have now added the following sentence to address this point: 'Rates of self-harm documented in general practice has been increasing for young people in the UK in the last two decades, especially in 13–16-year-olds.'

It would help if the authors were specific about the number of GP practices involved in the feasibility study.

This is a protocol manuscript and so we cannot say for certain how many practices will actually be involved (only what number we originally planned for). We mention in the methods of the manuscript that we aimed to recruit up to six practices. In the publication of the feasibility study findings the exact number of practices recruited will be stated. Thank you. We have reworded to 'to address'.

Any expected outcomes?

We are limited by the stipulated word count of the abstract for the journal but we do list intended primary and secondary study outcomes in the methods of the manuscript.

Plain English Summary

"Young people who self-harm experience further self-harm, anxiety and depression". Do the authors mean that young people who self-harm are more likely to repeat such behaviours in the future?

The costs of self-harming would be a better fit in the second paragraph as the first one presents and describes the problem of self-harming. Here the consequences and impact on the young people and their families would be more relevant.

We have reworded this sentence in response to this comment to the following as below: 'Young people who self-harm are more likely to self-harm again and suffer from anxiety and depression'

We have now moved the sentence about the costs of self-harm to the NHS to the second paragraph.

Introduction

The authors mention that 47% of those who self-harm have a mental illness. It would help if they were specific about which ones are related to self-harming. E.g., depression, eating disorders.

We have reworded this now to read 'rising to 47% in those with a mental health disorder such as anxiety or depression'. The national youth survey done in England from where this data comes from defined disorder according to ICD-10 categories.

The types of self-harming should be addressed in more depth and, especially some forms not widely known such as picking skin, hair pulling, which are very prevalent in young people. Then, it is important to discuss why young people might be engaging in self-harming behaviours.

We have now added in the second paragraph of the introduction the common methods of self-harm in young people noted in the literature such as skin pinching and hair pulling, and have described the functions of self-harm in 16-25 year olds.

The introduction would benefit from including specific trends in self-harming with regards to demographic characteristics (e.g., age groups, gender, ethnicity) and clinical features (e.g., depression, anxiety, eating disorders). For example, young people living in deprived areas, those with chronic mental health conditions such as anorexia nervosa, those who have lost a parent, and those in foster care might be presenting with a higher risk for self-harming. I think such social determinants are crucial when looking at self-harming in young people.

We agree with the reviewer that these social determinants are important in young people who have self-harmed. We have included one new sentence about the trends in rates of self-harm recorded in general practice in recent years which we feel add to the background of self-harm presentations in young people in general practice leading up to the rationale of this planned feasibility study. We have also added one sentence about likelihood of selfharming behaviour and socio-economic status and family structure in young people with a new reference from a 2017 Public Health England report.

The authors refer to GPs as having some skills to manage self-harm so I wonder about training in GPs in detecting and exploring self-harm. Do they receive adequate training to address and cope with such issues?

Thank you for this point. All qualified GPs should be competent in identifying and managing patients after self-harm in daily consultations – this is part of their core training to become GPs in the UK. However, the confidence of GPs varies in handling self-harm in young people in practice which is highlighted by past research (we have added one new reference to strengthen this point). This was a key factor in developing the COPING intervention: to improve GP confidence in handling self-harm in young people in practice.

Talking therapies might not be offering specific targeted interventions but self-harm is a symptom so the other risk factors in a young person's life may be addressed. We acknowledge that self-harm can be a result of underlying mental illness such as depression, anxiety, or post-traumatic stress, and so NHS talking therapy services may be beneficial, however this is not always the case. Young people may self-harm without meeting the diagnostic criteria for any mental health condition.

It would be helpful to explore more what elements of existing evidence-based interventions and approaches have informed the current intervention. What is the framework other than co-production? The co-production process could be described to show how this led to the development of COPING.

We thank the reviewer for this point. We have clarified that COPING was developed using a combination complex intervention development approach which brought together both evidence and theory and partnership approaches in the introduction.

We are preparing manuscripts about COPING's development and the co-production activity that occurred for submission to journals so we will not describe this in detail here. We do though under the 'intervention' section of the methods describe in detail the framework and methods that were used to develop COPING.

Methods: Eligibility criteria

How will the authors define self-harming?

We already state in the first sentence under eligibility criteria subheading that self-harm is defined according to the definition of NICE.

I wonder what is the role of carers in the process e.g., parents and how they can be involved at least in the qualitative phase of the study.

The design of this study was informed by carers of young people who have self-harmed. We have added this to the section about PPI in the manuscript to be explicit. We also state in this section that carers supported the development of COPING, and that they also are supporting the delivery of the feasibility study.

A table outlining the design and format of the intervention would be helpful and also the 8 behaviour change techniques and how these will be applied across the two

consultations.

While we agree that this may be helpful, we wish not to include this level of detail in this manuscript because these are intended to be described in the planned COPING-ID manuscripts which will be submitted shortly. In the second paragraph under 'Intervention' in the Methods we describe in detail the 8 behaviour change techniques and what happens in both appointments.

Patient and Public Involvement (PPI)

This section could include further information about the Steering Groups and how the content of the intervention was developed.

We have now added more information about the PPI groups in terms of numbers and composition and also how the study specific PPI group supported the development of COPING in the PPI section of the manuscript.

Competing Interests: No competing interests were disclosed.

Reviewer Report 23 August 2024

https://doi.org/10.3310/nihropenres.14737.r31763

© **2024 Rodham K.** This is an open access peer review report distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.



Karen Rodham

¹ Department of Health Psychology, University of Chichester, Chichester, England, UK ² Department of Health Psychology, University of Chichester, Chichester, England, UK

The researchers present a strong argument for a study to test the feasibility of an intervention that has been designed in collaboration with service users and GPs.

The suicide prevention in England strategy highlights the need for tailored support to young people who have self-harmed: a high-risk priority group. This proposed intervention developed by the research team could make an important contribution to this clear identified National need.

This protocol was clearly written. My job as a reviewer was made easy because of the care and attention that had been put into the protocol structure and construction. There was a strong narrative arc throughout and excellent use was made of signposting to keep the reader on track.

The only element of the document that could be improved in my opinion, is the plain English summary. There is much scope to simplify the language used here.

Rationale: The rationale for the intervention and for GPs to deliver the intervention was clearly and succinctly shared. The intervention itself has been co-created with collaborative input from

both GPs and service users. It now needs to be 'road-tested' to learn about the feasibility of embedding the intervention in GP practices.

Objectives: The objectives were clear and appropriate.

Methods: The methods were clearly explained, transparent and replicable. The team have employed recognised behaviour change models to inform the design of the intervention. Clear target behaviours have been detailed (the young person using new skills and the GP using the intervention in usual NHS practice). Data collection approaches and analysis measures are explained and a rationale offered for their inclusion.

PPI – has clearly been embedded in this proposal from the design of the original intervention, through to the proposed design of the feasibility study. Continued PPI is built into the proposal.

Risks have been considered and two risk protocols developed to ensure participants' safety and wellbeing.

Bigger picture – through PPI involvement and team expertise and experience, they have carefully considered the practicalities of maximising the possibility that this intervention will be deliverable in usual NHS practice.

In my opinion, this is an important proposal. It has been extremely well-presented and makes a strong case for the work to be supported.

Is the rationale for, and objectives of, the study clearly described? $\ensuremath{\mathsf{Yes}}$

Is the study design appropriate for the research question?

Yes

Are sufficient details of the methods provided to allow replication by others? $\ensuremath{\mathsf{Yes}}$

Are the datasets clearly presented in a useable and accessible format? Not applicable

Competing Interests: No competing interests were disclosed.

Reviewer Expertise: I have experience of researching young people who self-harm. I am a qualitative researcher with a track record of exploring how we can better help people to live well in spite of complex (multiple) physical health conditions.

I confirm that I have read this submission and believe that I have an appropriate level of expertise to confirm that it is of an acceptable scientific standard.

Author Response 10 Oct 2024

Faraz Mughal

We would like to thank the reviewer for their positive comments on our submission. In response to their suggestion, we have reworded the plain English summary to make the language used simpler, and so hopefully more readable for the public.

The plain English summary was not in the downloaded manuscript document so I have pasted it into the new submitted version after the scientific abstract.

Competing Interests: No competing interests were disclosed.